

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: Radium223_Review_revised_final_revised.doc).

Title: Clinical use of bone-targeting radiopharmaceuticals with focus on alpha-emitters

Author: Hinrich A. Wieder, Michael Lassmann, Martin Allen-Auerbach, Johannes Czernin and Ken Herrmann³

Name of Journal: *World Journal of Radiology*

ESPS Manuscript NO: 7637

The manuscript has been improved according to the suggestions of reviewers and the editor:

1 Revision has been made according to the suggestions of the reviewer

- (1) The language issues are edited and highlighted in the text.
- (2) Information on the production and availability of ²²³Radium chloride were added to the text (see page 6).

2 Revision has been made according to the suggestions of the reviewer

- (1) Comment WL1: A short running title is now provided (see page 1).
- (2) Comment WL2: The complete name of institution is now included in the text (see page 1).
- (3) Comment WL3-6: The post codes are now added (see page 1).
- (4) Comment WL7: The author contributions are now included in the text (see page 1)
- (5) Comment WL8-9: An abstract and a core tip are now added to the text (see page 3 and 4).
- (6) Comment WL10: The style of the number of the references is changed.
- (7) Comment WL11: The PubMed citation numbers for the reference list are now added (see references).

Thank you again for publishing our manuscript in the *World Journal of Radiology*.

Sincerely

Ken Herrmann